Author's response to reviews

Title: Clinical and Laboratory Findings Associated with Severe Scrub Typhus

Authors:

Dong-Min Kim (drongkim@chosun.ac.kr)
Seok Won Kim (chosunns@hanmail.net)
Seong-Hyung Choi (penta15@hanmail.net)

Version: 2 Date: 19 January 2010

Author's response to reviews: see over
Dear editor

Enclosing is a manuscript titled “Clinical and Laboratory Findings Associated with Severe Scrub Typhus (revised MS: 2812462873128525)”, which was electronically submitted for possible publication in original research paper section of BMC Infectious Diseases

This manuscript is new, is not being considered elsewhere, and reports new findings.

Our report highlights that clinician should be aware of the potential for severe scrub typhus, when scrub typhus patient was old age, or were presented with the absence of eschar, or such laboratory findings as WBC counts > 10,000/mm³, and serum albumin level ≤ 3.0 g/dL. Close observation and intensive care for scrub typhus patients with the potential for severe manifestation can prevent serious morbidity and mortality.

All authors have seen and approved the content and have contributed significantly to the work. And there are no potential conflicts of interest and financial supports for all authors.

Finally, we hope our results will be of great interest to physicians from western countries.

The corresponding author will be

Seok Won Kim
Department of Neurosurgery, School of Medicine, Chosun University,
588, Seosuk-dong, Dong-gu, Gwangju-city,
501-717, Korea.
Tel : (011) 82-62- 220-3126

FAX : (011)82-62- 227-4575

E-mail : chosunns@hanmail.net